European Institute for Crime Prevention and Control, affiliated with the United Nations (HEUNI) P.O.Box 444 FIN-00531 Helsinki Finland

Publication Series No. 61

Kauko Aromaa with

Minna Viuhko, Martti Lehti, Aleksey Taybakov, Petr Klemetshov Pavel Demin and Roman Tumanov

CORRUPTION ON THE FINNISH-RUSSIAN BORDER

Experiences and observations of Finnish and Russian civil servants and businesspersons on corruption on the border between Finland and Russia

Kauko Aromaa, Director European Institute for Crime Prevention and Control, affiliated with the United Nations, HEUNI

Martti Lehti, Researcher National Research Institute of Legal Policy, OPTULA

Petr Klemeshov, Head, Department (Republic of Karelia) of the Committee of Investigation of the General Prosecutor's Office of the Russian Federation

Roman Tumanov, Head Department of Civil Law Northern Branch (Petrozavodsk) of the Russian Legal Academy of the Ministry of Justice of the Russian Federation Minna Viuhko, Researcher European Institute for Crime Prevention and Control, affiliated with the United Nations, HEUNI

Aleksey Taybakov, Director Northern Branch (Petrozavodsk) of the Russian Legal Academy of the Ministry of Justice of the Russian Federation

Pavel Demin, Dean, Faculty of Further Education, Northern Branch (Petrozavodsk) of the Russian Legal Academy of the Ministry of Justice of the Russian Federation

Copies can be purchased from:

Academic Bookstore P.O. Box 128 FIN-00101 Helsinki Finland

Website: http://www.akateeminen.com

Criminal Justice Press P.O.Box 249, Monsey, NY 10952 USA

Website:

http://www.criminaljusticepress.com

ISBN 978-952-5333-75-6 ISSN 1237-4741

Printed by Hakapaino Oy, Helsinki, Finland

CONTENTS

1	IN	FRODUCTION	5			
2	ТН	E BACKGROUND AND OBJECTIVES OF THE STUDY	7			
	2.1	BACKGROUND	7			
	2.2	THE OBJECTIVES OF THE STUDY				
3	DATA AND METHODOLOGY					
	3.1	DATA COLLECTION	Ç			
	3.2	INTERVIEWS WITH REPRESENTATIVES OF THE AUTHORITIES				
	3.3	INTERVIEWS WITH BUSINESS REPRESENTATIVES	11			
	3.4	DISCUSSION OF METHODOLOGY	12			
4	CO	RRUPTION AS A PHENOMENON	15			
	4.1	DEFINING CORRUPTION	15			
	4.2	THE LIMITS OF CORRUPTION				
	4.3	THE ATTITUDES TOWARDS CORRUPTION	18			
	4.4	SUMMARY AND DISCUSSION	20			
5	CORRUPTION IN PRACTICE					
	5.1	EXPERIENCES OF CIVIL SERVANTS WITH CORRUPTION	22			
	5.2	EXPERIENCES OF BUSINESSPERSONS WITH CORRUPTION				
	5.3	SUMMARY AND DISCUSSION	25			
6	COOPERATION BETWEEN THE FINNISH AND RUSSIAN AUTHORITIES					
	6.1	OBSERVATIONS OF REPRESENTATIVES OF THE AUTHORITIES CONCERNING				
		COOPERATION BETWEEN THE FINNISH AND RUSSIAN AUTHORITIES	27			
	6.2	OBSERVATIONS OF BUSINESS RESPONDENTS ON THE ACTIVITIES				
	<i>c</i> 2	OF THE AUTHORITIES				
	6.3	SUMMARY AND DISCUSSION				
7	RE	GULATION AND PREVENTION OF CORRUPTION	32			
	7.1	CORRUPTION-RELATED TRAINING, GUIDELINES, AND FAMILIARITY				
		WITH LEGISLATION				
	7.2	THE PREVENTION OF CORRUPTION				
	7.3	SUMMARY AND DISCUSSION	37			
8	CO	RRUPTION ON THE FINNISH-RUSSIAN BORDER	39			
		WHITE LIMIT IGED IN THE ODICINAL BININGH AND DUGGLAN COVE	NICED X7			
	ITEKA FPORT	TURE UTILISED IN THE ORIGINAL FINNISH AND RUSSIAN COUL	NTRY 46			

1 INTRODUCTION

According to Transparency International, the Finnish-Russian border represents one of the sharpest corruption borders in the world (cf. the home page of Transparency Finland). For this reason, research on corruption on the Finnish-Russian border is of particular interest and also of particular importance. There is only fragmentary knowledge concerning this phenomenon, and even this is largely based on hearsay and informal experience. There are, for example, different kinds of beliefs about corruption in commerce between Finland and Russia and in the relationships between the Russian and Finnish authorities, but these beliefs are not based on systematic research on the topic.

This research project was carried out on the initiative of the Finnish Ministry of Justice. Its objective has been to provide the relevant authorities with new and better knowledge regarding corruption on the Finnish-Russian border. The results can also be used as working material at joint seminars and training events. This is expected to facilitate constructive debate between the authorities of both countries, based on more concrete observations than what has been available previously.

The Finnish part of the project was carried out by the European Institute for Crime Prevention and Control, affiliated with the United Nations (HEUNI), in cooperation with the National Research Institute of Legal Policy. The study approached Finnish civil servants and representatives of Finnish enterprises active in Russia, in order to shed light on their observations and experiences of corruption, corruption prevention, and cooperation among authorities on the Finnish-Russian border and in cross-border activities. Furthermore, the study has looked at the experiences of civil servants and of employees of business enterprises about corruption on the border, as well as other problems met in cross-border activity.

Parallel to the Finnish study, a similar study was carried out in Russia (Republic of Karelia) by the Northern Branch of the Russian Legal Academy of the Ministry of Justice of the Russian Federation. This study targeted representatives of the Russian authorities and of Russian business enterprises active in Finland, focusing on the same topics as the Finnish part of the project.

The focus of the project is on corruption specifically on the Finnish-Russian border, and in particular on the border between Finland and the Republic of Karelia. Consequently, the study does not seek to grasp the full picture of the phenomenon of corruption in either Finland or Russia. This joint report summarises the central findings of both studies. The full report of the Finnish

study is published separately. The Russian report was received as a manuscript. 2

The structure of this report is as follows. Chapter 2 explains the background and objectives of the study. Chapter 3 presents the research methodology and research data. Chapters 4–7 present the main results of the study. First, Chapter 4 analyses the issue of how the respondents understood the phenomenon of corruption and how they felt about corruption. Next, Chapter 5 discusses the personal experiences that the interviewed representatives of the authorities and business enterprises disclosed with regard to border corruption. Chapter 6 deals with cooperation and activities by the authorities that relate to border corruption. Chapter 7 presents the respondents' observations and opinions about corruption control and prevention. Finally, Chapter 8 summarises the conclusions and recommendations of the study.

¹ Viuhko, Minna, Lehti, Martti (2009). Korruptio Suomen ja Venäjän rajalla. Suomalaisten viranomaisten edustajien ja liikemiesten kokemuksia ja näkemyksiä korruptiosta itärajalla. Oikeuspoliittisen tutkimuslaitoksen tutkimustiedonantoja 99. HEUNI Publication Series No. 60.

² Taybakov, A.A., Demin, P.I., Tumanov, R.V., Klemeshov, P.A. (2009). Corruption on Russian-Finnish border. Comparative analysis in Finland and the Republic of Karelia, Russia. The Northern Branch (Petrozavodsk) of the Russian Legal Academy of the Ministry of Justice of the Russian Federation. Unpublished manuscript.

2 THE BACKGROUND AND OBJECTIVES OF THE STUDY

2.1 Background

Corruption prevention was included as a new topic in the 2007–2008 cooperation programme between the Finnish and the Russian Ministries of Justice. For the cooperation programme it was agreed to organise two seminars and one study tour for the Russian authorities. When the cooperation events were being organised, it became evident that there is hardly any research-based knowledge about corruption issues on the Finnish-Russian border.

Accordingly, a research project was organised in 2008–2009 upon the initiative of the Finnish Ministry of Justice. The objective of the project was to obtain concrete and timely information concerning corruption on the Finnish-Russian border, both from the Finnish and from the Russian perspective. The project was carried out in cooperation with the Northern Branch of the Russian Legal Academy of the Ministry of Justice of the Russian Federation. The main objective was to provide the authorities with information about corruption on the Finnish-Russian border that would be more reliable than what has been available to date and that would be derived from several sources. A further objective was that the results would be available for use as discussion materials for joint seminars and training events, and in this way facilitate constructive dialogue between the Finnish and Russian authorities.

The current research project was intended to provide research-based knowledge to replace impression-based "knowledge". Furthermore, it is important for authorities to receive first-hand knowledge about what kinds of corruption-related problems have been experienced on the Finnish-Russian border, what are the attitudes towards such problems and what kinds of attempts have been made to solve them, and eventually, how the trust and cooperation between the different actors on the border could be increased.

2.2 The objectives of the study

One of the objectives of the study was to analyse attitudes related to corruption, and how corruption is understood and defined in Finland and in Russia. The idea was also to analyse whether there are differences in this regard between authorities and business representatives. A second central objective was to collect timely information about the (possible) personal experiences that the research respondents had about corruption on the Finnish-Russian border and in cross-border activity. One further point of interest was whether there have been similar experiences in different business sectors, what harm corruption does to business, and whether there have been initiatives to prevent or reduce

such harm. Further topics of the study were corruption prevention and cooperation between the Finnish and the Russian authorities.

For both studies, the Finnish partner prepared detailed research guidelines that were also discussed at joint meetings. The questions to be studied were:

- 1) What is corruption as understood by representatives of the authorities and of businesses?
- 2) What particular phenomena are linked to corruption?
- 3) What are the limits of corruption? What is understood to be allowed and what is not?
- 4) What kinds of corruption-related problems have the respondents come to know about in practice?
- 5) What kinds of corruption-related problems have been observed on the border and in cross-border cooperation in practice (if any)?
- 6) What is known about the regulation of corruption (both in Finland and in Russia)?
- 7) How do the representatives of the authorities see corruption prevention and their own role in prevention?
- 8) How do the representatives of the authorities perceive Finnish-Russian cooperation, and what possible problems do they see as being related to this cooperation?
- 9) What is the opinion of representatives of business enterprises about the effectiveness of the activities of the authorities and corruption prevention?

The Finnish study targeted police, customs and border guard authorities, and business enterprises that were active across the Finnish-Russian border. The research material was collected in qualitative thematic interviews that were systematically recorded.

In the Republic of Karelia, the study was carried out in a similar fashion. The target groups of the study were defined in an analogous fashion as in Finland. However, because of differences in the authority structures, the selected target authorities were not identical to the Finnish case, because on the Russian side, some special authorities exist that have no direct counterpart in Finland. The starting point was that the fieldwork was to be carried out in the same way as in the Finnish study. However, deviations from this became necessary, since personal interviews turned out to be impossible in many cases (see Chapter 3, "Data and methodology").

3 DATA AND METHODOLOGY

3.1 Data collection

Separate interview topic lists were developed for the interviews with the representatives of the authorities, and with the representatives of businesses. They were partly overlapping, with a number of the same questions for both respondent categories, but there were also some questions designed for each respondent group separately (for example, the questions concerning the professional history of the respondents and, for business respondents, the business sector of the company). Otherwise, the questions dealt with the topics defined above in section 2.2.

The Finnish data were collected in two parts. Researcher Minna Viuhko conducted the interviews with representatives of the authorities between late November 2008 and the end of January 2009. Researcher Martti Lehti conducted the business representative interviews in January-February 2009. The interviews were qualitative thematic interviews. In the interview situation, the researcher met with one respondent at a time, with the exception of three business interviews in which two company representatives participated simultaneously. In the interviews, the topic list was followed in a flexible fashion so that the questions were not always presented in an identical order. The interview began with the researcher asking the respondent to describe his/her work and his/her work history. After these background questions, the researcher asked questions related to the definition of corruption before the interview moved on to questions about possible concrete cases and experiences of corruption.

All the Finnish interviews were recorded and transcribed. Each respondent was asked for his or her permission to record the interview, and all of them agreed to this.

Also in Russia, the fieldwork was made in two parts. The interviews with civil servants were carried out between early December 2008 and the end of February 2009. The interviews of business representatives were conducted from early January to early March 2009. In the Russian study, the interviews could not be recorded at all due to the refusal of the respondents to allow this. Five unrecorded interviews were conducted with civil servants. In the other cases, the civil servant respondents refused to be interviewed, and were asked to complete a written questionnaire. The business representatives refused to be interviewed but agreed to complete an e-mail questionnaire.

The possible implications of these differences in data collection are discussed below in section 3.4.

3.2 Interviews with representatives of the authorities

In both countries, the target was set at 15 interviews of civil servants.

In Finland, the final data comprised interviews with five representatives of the police, six customs officers, and four border guard representatives. The interviews were conducted between the end of November 2008 and the end of January 2009. Their length varied between 30 minutes and three hours.

The respondents were recruited by various methods. First, the researcher listed persons employed in the three sectors, whose work was in some way related to the cooperation between Finland and Russia, in particular the Republic of Karelia. The choice of respondents was then made by prioritising persons who had expertise in issues related to corruption. The selected persons were contacted by telephone and e-mail. Most of the contacted persons agreed to be interviewed. Some of the potential respondents also recommended a colleague to be interviewed in his/her place or in addition to him/herself. The researcher also asked during the interviews who else the respondent thought should be interviewed.

The contents and the information yielded by the interviews varied considerably depending, for example, on the length of the respondent's work experience or his/her occupation. Some had much to tell, while others were quite laconic, saying they had no experience or opinions about corruption, or even about cooperation between the Finnish and Russian authorities. It is relevant to note that the topic was quite delicate and challenging. Some respondents did not want to talk about corruption even on a general level, or they said that they had no knowledge of the topic whatsoever. Many respondents also felt that it was difficult to define corruption as such. Some of them may also have felt uneasy about discussing corruption-related issues, whether referencing the Russian or Finnish authorities. On several occasions, the respondents seemed to be quite reserved, and chose their words very carefully. This may in part be because the respondents did not want to jeopardise cooperation between the two countries, and were reluctant to disclose information or opinions that the other party might interpret as being insulting. On the other hand, some respondents were very talkative and ready to speak about specific corruption-related incidents and about problems in cooperation between the authorities.

In Russia, the data relating to the authorities were harder to come by. The set of respondents comprised six customs officers, seven representatives of the administration of the federal service on veterinary and phytosanitary control, one representative of the administration of the federal service of officers of justice, and one representative of the department of international cross-border cooperation and protocol of the Ministry of Economic Development. The respondents were identified after formal applications for official permission from each of the relevant authorities. It was at this stage that the border administration declined to participate in the study. Many of the civil servants

who were approached refused or recommended their colleagues. Only five people agreed to be interviewed, but without recording. All the others agreed only to completing an anonymous questionnaire.

The contents and the value of information directly depended on the legal education, length of work experience and occupational position of the respondent. Respondents with longer work experience gave more detailed answers and also answered all of the questions thoroughly. Other less-informed respondents usually claimed no knowledge of the issues in question.

3.3 Interviews with business representatives

The target was to obtain 15 business representative interviews in both countries.

In Finland, the arranging of the business representative interviews was initiated by developing a list of potential targets. This was achieved by asking the Finnish-Russian Chamber of Commerce and Finnish enterprise authorities and chambers of commerce for information about companies established in Karelia or operating in Karelian territory. Of these sources, the Finnish-Russian Chamber of Commerce proved to be the most useful one, providing a list of more than ten member companies. Furthermore, the Internet was searched for participants at business seminars related to trade with Karelia and for lorry companies engaged in road transport to Russia. Suggestions for potential contacts were also asked in the interviews with representatives of the authorities.

From these sources, a list of 50 companies was compiled, and these were contacted by telephone. The selection criteria were that the travelling distance from Helsinki to the company office should not be more than 350 kilometres, and that the respondents should be persons who had extensive (at least 5 years) experience with trade with Russia. The target was to locate a group of 15 companies engaged in exporting, importing, production, or transport.

Eventually, interviews were completed with 13 persons representing ten companies. In addition, one business consultant was interviewed in connection with the interviews with representatives of the authorities. The interviews were conducted in January and February 2009, and their length varied between 40 minutes and one and a half hours. Because more than half a year passed between the interviews and the publication of the research report, some of the respondents were contacted again in July 2009. The objective was to find out whether, from the perspective of the company, the situation had changed since the end of February.

The majority of the respondents had at least ten years of experience with trade with Russia, and one-third had also lived in Russia for extensive periods of time. All respondents were Finnish nationals. The companies comprised both

Finnish and foreign-owned businesses; however, none of them were Russian-owned.

When contacting the businesses, transport companies and timber import companies indicated that they were particularly interested in the study. Since the objective was to locate information about phenomena that the companies felt to be problematic, it was decided that companies that were reluctant to participate should not be pushed to do so. Of the companies that were contacted, four had terminated their activity in Russia in the course of the previous year, five refused to participate, and six did not respond to repeated contact requests.

Two observations deserve attention in this context. First, the participation of only one representative of the export trade could be secured. It is possible that export businesses felt that the topic was unfamiliar or irrelevant to them. However, it is also possible that the topic is more delicate for export businesses. In corruption related to the export trade, the role of the business is active and predatory, in contrast to, for example, corruption targeting the import trade and the transport sector, where the businesses are more clearly in the role of passive victims. Also, financial interests are likely to be larger in corruption that is – possibly – related to export trade. The second relevant observation was that both of the two companies with a Russian background flatly refused to participate in the study. The persons who were approached in the companies with a Russian background were Finnish nationals. However, since the number of such companies approached was only two, this may of course also be explained by mere chance.

The selection of respondents for the Russian business survey was made by approaching the Industrial Business and Trade Association of the Republic of Karelia. From this source, information was received about 34 entrepreneurs and companies that had activity connected with Finland. Many potential respondents refused to participate, referring to business and commercial secrets. Eventually, 15 companies agreed to participate. They represented small and medium-size businesses, and their activity was connected with the timber trade and the transport of goods and people. One respondent represented a large-scale company.

All respondents agreed to answer only anonymously and by e-mail. They were encouraged to contact the research team and ask for clarification, and they often used this opportunity. It is the impression of the researchers that all of the respondents treated the questions responsibly and were interested in the results of the study.

3.4 Discussion of methodology

Since the samples are not random and the research approach is qualitative, the results cannot be generalised in a statistical sense to correspond to all relevant authorities and companies in Finland and Russia. The results provide

information about the attitudes and experiences of the selected persons. The qualitative approach is understood to be a good tool for illuminating a phenomenon that has not been studied much and about which there are only rather vague ideas and subjective, often prejudice-based knowledge. The qualitative approach is also suitable for research that attempts to study how a given phenomenon is generally understood and what people mean by it. It is also relevant to observe that certain issues emerged from several interviews, making it likely that the observation in question is not merely the view of a single person but is representative more generally of the group (such as the authorities operating on the border) from which the respondents are selected.

It may be relevant to ask how good a research method it is overall to ask representatives of the authorities about corruption related to the authorities. Even in cases where the interviewee does know about corruption issues related to his or her own organisation or to the corresponding organisation on the other side of the border, it is possible or even likely that the respondent does not want (or would even dare) to talk about it. With regard to one's own organisation, to talk about problems and unlawful behaviour may often not be considered to be acceptable even if the person himself or herself has not been involved. Sometimes there may be fears of reprisals or other unpleasant consequences within the organisation even if the interview is anonymous. There may also be the feeling that it is not desirable to speak openly about problems in the neighbouring country either. Negative views may be thought to damage cross-border cooperation with colleagues on the other side. This cooperation and confidential relations with the colleagues in the neighbouring country are necessary for the authorities to be able to carry out their duties successfully.

The Finnish business representatives talked more openly than civil servants, and came forward with many critical observations. They reported many corruption-related problems that the other respondents – both Finnish and Russian – remained rather vague about. This would imply that the Finnish businesspersons saw the study as a new opportunity to get attention to their everyday problems, with regard to which they felt they had not received sufficient support from the Finnish authorities and government.

The replies of the Russian businesspersons were rather brief and did not go into very much detail. This is likely to be a consequence of the survey mode (anonymous e-mail questionnaire) which had to be applied in this case because the respondents categorically refused to be interviewed in person. Written replies may be seen as potentially dangerous, but also as simply too much work. What is most consequential is the fact that the questions were designed for use in a qualitative face-to-face interview, and not for use as a written questionnaire. In a face-to-face interview, numerous follow-up questions are typically used for increased detail. This cannot be done if the survey is made with a short written questionnaire that comprises only the general points of the study.

In a comparative perspective, it is problematic that the two studies were not identical. They were not identical with regard to the data collection methods, nor were they identical with regards to which sectors the survey respondents represent. Furthermore, the results were not reported in a consistent manner for both countries. Some of the interview questions also seem to have been misunderstood by the Russian team. This has given rise to some difficulty in the compilation of a joint research report. This problem would certainly have been much alleviated if it had been possible to achieve more systematic and intensive cooperation between the two research teams, including a common working language. In this case, due to time and resource constraints, this was not done. Should similar cooperation continue, such issues would need more attention.

A classic question concerning the applied methodology is whether the results represent the "truth". If somebody tells an anecdote to the interviewer, this is of course just an anecdote or a story that the respondent chooses to tell. The qualitative interview and its variants are in the first place able to inform the researcher about how the respondents understand the research topic, and how they see the topic. The stories they are telling about observations related to the topic represent anecdotal evidence that is not equivalent to proven facts, and they do not constitute evidence that would hold in court. They do, however, unveil culturally common and shared observations that become more convincing if a large proportion of the target group corroborates them independently of one another. As explained above, this particular method was chosen because there was little systematic and documented knowledge about the phenomenon in question, and such a study was deemed to be likely to provide new insight into how corruption is perceived among the people comprised in the target groups. A typical case of added value is when the results differ from expectations formulated on the basis of earlier work, thereby informing the researcher that the frame of reference needs to be adjusted and complemented with insights derived from the qualitative work.

4 CORRUPTION AS A PHENOMENON

4.1 Defining corruption

Finland

In Finland, many representatives of the authorities had difficulties in defining corruption. Some respondents were not able to answer anything at all to this question. Most of them, however, did have some kind of a general idea of a definition. The majority defined corruption to be, in the first place, equivalent to bribery. However, they did understand that, in addition to bribes, any kind of abuse of a position of authority for benefit was illegal. The views of businesspersons corresponded closely to the Finnish Criminal Code description of the offence of giving a bribe. In both groups, a bribe was understood more broadly than as just a financial payment, to include any other kinds of direct or indirect material or immaterial benefits.

With regard to different possible constellations of corruption, the representatives of both the authorities and businesses focused on corruption between authorities and businesses. Corruption located solely in the private sector did not emerge as a topic in the interviews.

Many respondents distinguished large-scale from small-scale corruption. In large-scale corruption, large sums of money were involved, while small-scale corruption was of an everyday nature, such as giving a pack of cigarettes, a chocolate box, or a small amount of cash in exchange for flexible service by the authorities. Both types of corruption were seen to go together. Additionally, no difference was seen in the reproachability or harmfulness of either. The Finnish respondents, both the representatives of the authorities and businesspersons, felt that everyday small-time corruption was illegal and unacceptable in the same way as grand corruption.

In several interviews, corruption was explicitly stated to exist "on the other side of the border", or it was said to "be part of that culture". Both the representatives of the authorities and the businesspersons generally thought that corruption is closely related to the Russian authority culture (bureaucracy). Even so, some businesspersons with broad international experience observed that in an international perspective, the relatively low corruption of the Finnish authorities is the exception, not the corruption in Russia.

The businesspersons as well as some of the representatives of the authorities were of the opinion that the limits of corruption and the rules of the game regarding corruption are dissimilar in the two countries. According to the representatives of the Finnish authorities, for instance, bribery in the civil service is always strictly condemned in Finland, while small-scale corruption in Russia was seen as the "national custom".

Overall, both groups had a largely convergent view about corruption, and this view was largely congruent with the definition in the Finnish Criminal Code. Corruption was understood to comprise rather broadly different phenomena in the grey economy, as well as the discretion exercised by the authorities. The examples that were referred to would usually be classified as crimes according to Finnish and Russian law alike.

Russia

In the Russian data, most civil servants defined corruption as, for example, abuse of an official position for one's own interest, or the granting benefits in exchange for money, or receiving benefits that are not always in the form of money. They also referred to situations where government decisions or the decisions of an authority can be bought. Corruption was also seen to comprise cases where a representative of an authority neglects his or her duties, or conceals violations. Another example of corruption was if civil servants rendered exceptional services, or reduced regular fees.

The Russian businesspersons had slightly more general ideas about corruption. Many saw corruption as a way of solving problems with the authorities in order to get the "right" kind of decision. Another common notion was that corruption refers to the bribing of the authorities as well as the blackmailing of entrepreneurs by the authorities. One respondent said that corruption means a relationship between the authorities and business where both benefit. One other respondent saw corruption as a socially dangerous phenomenon, where officials deliberately use their position in exchange for (material) benefits. All business respondents connected corruption of state authorities and corruption in business with the bribing of officials or with the blackmailing of businesspersons by the authorities who have the power to assist in solving their problems.

4.2 The limits of corruption

Finland

According to the civil servants who were interviewed, the borderline between gifts and bribes was vague, and this was also understood to depend on the situation. The formal position of the recipient (director general vs. lower official, or individual official vs. a delegation) as well as at which stage the gift is received (e.g. before or after negotiations), together with the motive for giving the gift were seen to be relevant when the limits of acceptability are assessed.

A gift was understood to be an expression of politeness, friendship or goodwill, nothing more. A standard working lunch, a bag of coffee or a small souvenir

were regarded as acceptable gifts. The basic requirement was that they must not be of a high value.

The representatives of the authorities commented generally that it has become less common in Finland over time to give any gifts to the authorities, and Finnish authority practice has become increasingly strict about accepting different kinds of hospitality and other benefits. However, they also took notice of the fact that the limits on and the attitude towards gifts vary internationally. Some respondents commented that in Russia, "nothing works without presents".

In this respect, the businesspersons had views that were very similar to those of the civil servants. Several respondents underscored that in Russia, the limits between acceptable and illegal influence in respect to the authorities are more flexible than in Finland. It was said that, culturally, it was more common in Russia than in Finland for businesspersons to give gifts to the authorities with whom they cooperate, and the gifts are more valuable. Gifts and active signs of cooperation are also directly expected in Russia, which is different from what is the case in Finland. At the same time, the businesspersons commented that also in Russia, the line between bribes and gifts is clear, both for the giver and the recipient.

The difficulty in distinguishing the limits to corruption was reflected in the problem that when respondents spoke about gifts, presents and bribes, it was not always clear whether reference was being made to a bribe in the sense used in the Criminal Code, to a morally questionable "present" or to an acceptable gift. If a Finnish respondent says that a Russian does not see the giving and receiving of presents as being corruption, is he/she then himself/herself speaking of a gift in the sense of a bribe? Further, is such a gift or bribe considered to be acceptable if there is no other alternative? In other words, in cross-border activity should one apply the cultural norms of the person who is giving the benefit or of the one person who is receiving it? In cases where the action is unacceptable according to both sets of cultural norms there is no problem, but the grey area between clearly criminal and clearly legal behaviour may often be broad.

Russia

The civil servants thought in general that corruption in civil service is not acceptable. However, one respondent was of the opinion that the limits are set by one's conscience. In his view, it is thus not prohibited to express gratitude in the form of small presents, such as flowers to the amount of 3000 roubles (approx. 66 euros).

Four-fifths, or a clear majority of the interviewed businesspersons said that it is not allowed to give presents to officials. A few respondents said that small presents or souvenirs are acceptable.

Thus, the Russian views quite clearly reject the giving of presents, with a few exceptions where small gifts are seen to be unproblematic. There is not a large difference between the views of representatives of the authorities and of businesspersons. It is interesting to notice that the Russian respondents (both civil servants and businesspersons) seemed to define the limits of acceptable gifts contra corruptive presents in Russia more narrowly than did the Finnish respondents.

4.3 The attitudes towards corruption

Finland

Finnish civil servants said that their attitude towards corruption was extremely negative. For them, corruption and bribery were not acceptable under any circumstances, and in particular not in the work of civil servants. They also thought that any corruption cases that come to light must always be dealt with very severely.

According to the representatives of the Finnish authorities, the Russians also know that Finnish civil servants cannot be bribed, and attempts to bribe them are now very rare. Some respondents said that Russians often think that Finnish civil servants are stupid and simple-minded since they do not understand how to use their position to their own benefit.

The civil servants had different kinds of opinions about corruption in Russia. Some maintained that Russia is very corrupt, and that there is a lot of money in corruption. Others thought that the Russians themselves do not consider "speed money" and the principle of mutually beneficial arrangements to be corruption or reproachable, and because of this it cannot be judged as corruption by the Finns either. According to this view, the problem is a cultural one rather than a criminal one.

Some representatives of the authorities also maintained that in some situations in Russia, Finnish citizens were not given any alternative other than to pay what was required. Such situations were said to be particularly relevant for employees of business enterprises. Consequently, the differences between Finland and Russia in public sector corruption were to a large extent also seen as a cultural difference. Unlike their Finnish colleagues, Russian civil servants were thought to consider that it is acceptable to ask for a fee or a service in exchange for carrying out their official duties.

Most of the interviewed representatives of the Finnish authorities thought that Russian society was fundamentally different from Finnish society: friendship and survival networks were thought to be a necessity for a Russian, in order for everyday life to work out at all.

The different attitudes towards corruption were understood to result partly from differences in incomes and the standard of living and social structures.

Explanations were also seen in whether society was seen as being a just one for everybody, and whether there was high or low confidence in the authorities. The example was cited that the confidence of the public in the police is at a totally different level in Finland than in Russia. The Russians' distrust of the authorities was seen to promote corruption.

According to the representatives of the authorities, corruption was also supported by the complicated structure of the Russian government and its conflicting jurisdictional relationships. They considered that the "endless bureaucracy" in Russia has created the situation where it was necessary to use various methods in order to speed up bureaucratic procedures.

The Finnish businesspersons had a more pragmatic attitude towards corruption in Russia than did the civil servants. Their attitude was also basically negative, but this was in part because corruption was considered to be something that was causing harm to business. The more experience the respondent had with operating in different kinds of cultures, the more complex was his/her understanding of corruption as a phenomenon as well. The businesspersons saw corruption as an acceptable practical method of solving problems. To participate in corruption, to "play by the local rules" was explained to be necessary in business directed at Russia. In the cases given by the businesspersons as examples, the corruption almost always involved the company submitting to illegal requests made by the Russian authorities, not cases in which the companies themselves would have taken the initiative, corrupting the authorities for their own gain.

Nevertheless, some respondents also described instances where they did try to corrupt authorities on their own initiative. For example, businesspersons had, in connection with court cases, attempted to influence the court's decision by a bribe. The respondents were of the opinion that also this was justified in some special cases. They regarded themselves as the victims of corrupt practices rather than as criminals.

Russia

Half of the interviewed Russian civil servants said that corruption cannot be justified under any circumstances. However, others thought that there are situations where corruption is acceptable. Such situations were connected with the weak social protection of civil servants and their family members. One example of such situations was that "when a person is being influenced, blackmailed, threatened, or his family members are being threatened, he may use his official position to pass information to third persons". Another example was that "if a close relative is seriously ill and money is needed for an expensive operation, such a case can be understood on humanitarian grounds". A further example of an acceptable situation was "if a person cannot afford to feed his children, or his close relative is seriously ill, or all his property has been burned in a fire, and in such a situation a concerned party offers money".

All three examples refer to quite extreme situations of distress, not to everyday routine requests for fees to be paid for services.

With one exception, all of the Russian business respondents replied that there is no situation in which corruption would be acceptable. The one exception was a respondent who stated that such a situation might occur if corruption is needed to save one's own business.

Both the Russian civil servants and the Russian business respondents gave only very brief comments on this issue. This may very well be a consequence of the survey mode: written answers tend to be more careful and shorter than answers given in a face-to-face situation where follow-up questions can also be used to obtain more complete answers.

In any case, the replies of the Russian respondents as to whether corruption could be justified seem to imply that they were thinking of exceptional and "serious" corruption rather than everyday "speed money" situations. The written question referred to "corruption", not to "presents" or other terms that might fall in the grey zone between acceptable gifts and clear-cut bribery offences. On the other hand, as mentioned in the previous chapter, the Russian respondents (both civil servants and businesspersons) seemed to define the grey zone between acceptable gifts and illegal bribes very strictly, even more narrowly than the Finnish respondents.

When discussing explanations for what gives rise to corruption, the Russian civil servants thought along lines similar to those of their Finnish colleagues. They said that corruption is created by the government itself by the unclear drafting of normative and legal acts, and the complicated procedures for preparing documents. Furthermore, corruption is facilitated by the lack of civil control of state institutions, together with the lack of a system of timely investigation of complaints from citizens, and the limited access that citizens have to state institutions. A final contributing factor was said to be the fact that officials often cover up one another's illegal behaviour.

4.4 Summary and discussion

Finnish civil servants and businesspersons understood corruption basically in a very similar way. Both rejected corruption as a phenomenon. However, some of the business respondents did not consider situations where one agreed to pay bribes on request as being particularly reproachable when doing business in Russia since this was in their view a necessity.

The respondents' perceptions about corruption were strongly influenced by their extensive personal experiences in Russia. In their view, it is currently impossible to be involved in business or to work with the authorities in Russia without coming into contact with at least some issues related to corruption.

A particular dilemma, identified by both the Finnish civil servants and the Finnish business respondents, was the difference between the two countries in

the sociocultural norms related to corruption. In the Finnish view, the limits of corruption in Russia were not identical to those in Finland. In cross-border activity, this became a problem: should one behave according to the Russian or the Finnish standards? The answer of the businesspersons was to follow the Russian sociocultural norms. This became a problem when doing so constituted an offence according to Russian criminal law also. Even in such a case, the understanding among businesspersons was that paying a bribe on demand was sometimes the prerequisite for being able to operate at all, i.e. being a party to corruption was a necessary evil. The Finnish civil servants were less flexible about this, perhaps since they were able to comment on the question on an abstract level, while for the businesspersons the problem was a matter of their livelihood.

The Russian responses seemed to focus more clearly on "real" or large-scale corruption rather than on everyday small-scale "speed money" collection. On this level, their attitude towards corruption was clearly negative. However, the comments by civil servants may be interpreted to refer also to small-scale money collection when they said that corruption is created by the government itself by the unclear drafting of normative and legal acts, the complicated procedures for preparing documents, and the lack of a system for the timely investigation of citizen complaints. It is unclear why they did not discuss this more openly since it is hard to believe that they did not know about common practices. The study itself provides no clue as to how to settle this issue. Thus, it is only on a speculative basis that we may suggest as one possibility that the Russian civil servants did not talk more openly because they wanted to protect their colleagues or themselves, together with the speed money culture in general.

The Russian responses also indicated that the attitudes of Russian civil servants and businesspersons towards corruption, the "rules of the game", may be more ambiguous than many Finnish businesspersons and civil servants tend to think.

5 CORRUPTION IN PRACTICE

5.1 Experiences of civil servants with corruption

Finland

The respondents were asked about concrete experiences with corruption, either personal cases, cases that occurred in their working environment, or other generally known cases. The focus was on corruption related to the activity of the authorities and business, in particular to events on the border.

The respondents told only about a few bribery attempts. These were individual cases, and the persons offering bribes were individual citizens. The incidents involved small amounts of money, and the matters concerned situations where a person had been fined or turned back at the border. The person in question had offered money in order to have the measure waived or in order to take care of the matter "on the spot". The civil servants said they had never accepted such offers. Over the recent years, such incidents were said to have decreased but at least three respondents had personal experience with such situations. In two cases, the bribe was offered by a Russian, and in one by a Finnish citizen. In addition, there had been incidents at passport control where a banknote had been inserted inside the passport. In these cases, the person crossing the border was simply informed that "in Finland, you don't need to pay".

No other examples of bribery were found in the interviews. Some respondents commented that cases might easily remain uncovered. Some of the representatives of the authorities were very careful about what they said, and they did not seem to like discussing the matter even in very general terms. Of the different representatives of the authorities, the customs officials talked more openly about corruption and problems related to corruption, and gave concrete examples.

Some of the respondents underscored that the difference between Finland and Russia has diminished over the recent years. Some civil servants said that it is unrealistic to believe that there is no corruption at all in Finland. In their view, corruption in Finland is mainly to be found in the private sector. In particular, double invoicing was mentioned in several interviews as an example of business-related corruption. (Double invoicing along the Finnish-Russian border generally involves declaring goods at their true value when exporting them from Finland, but then declaring them at a much lower value on the other side of the border, when importing them into Russia. The benefit lies in the fact that the import duties are imposed on the basis of this artificially lower value.)

Russia

All of the Russian respondents stated that they had no first-hand information about corruption in cross-border cooperation between Russia and Finland. Basically, what they know about corruption comes from media and rumours, and these are mostly about corruption violations involving the customs authorities of the Russian Federation.

Most of the Russian civil servants had never actually seen any indications of corruption cases in their working environment although they did not exclude the possibility that such cases may exist. One respondent, however, did point out that corruption was to be found in the customs during the 1990s. In his view, it was easy at that time to exploit the imperfections of the law and constant regulatory changes.

Two civil servants stated that they had been offered bribes but had rejected these because of the serious consequences. Most of the respondents thought they had not been offered bribes because of their low official positions. They also explained that usually corruption cases, if detected, are kept secret.

The civil servants said that most corruption cases in Russia are connected with the border and arise at points in transit at customs control, especially when crossing the border is simplified for timber transport and large-scale consignments. According to these civil servants, corruption is still possible in Russia today, with the customs authorities, in the migration service, in travel companies, at the Finnish consulate, and among Finnish entrepreneurs.

As to the Russian situation generally, most Russian respondents thought that corruption is a mass phenomenon. They believed that corruption in Finland is possible only among individual civil servants, citizens and companies, and that corruption on the ordinary level is practically impossible and strictly suppressed. In this view, one respondent suggested that joint anti-corruption measures would promote cooperation between the authorities, the exchange of experiences and the detection and apprehension of lawbreakers. Such cooperation would provide a useful learning experience.

5.2 Experiences of businesspersons with corruption

Finland

None of the Finnish businesspersons told about experiences of corruption involving the Finnish authorities. The cases of corruption in Finland were all related to customs or tax fraud in trade from Finland to Russia (such as double invoicing), where Finnish export companies and forwarding agencies were

involved, or to corruption connected with crimes in the Russia to Finland direction (such as the illegal fuel trade).

The Finnish businesspersons had a more direct relationship to Russian corruption than the civil servants. Almost all respondents had first-hand experiences with corruption in Russia. Mostly, these concerned small-scale corruption where the businessman was in the role of a victim who must consent to various additional illegal or pseudo-legal fees for permits, licences and services, or illegal or pseudo-legal fines for (alleged) misdemeanours. The individual amounts of money involved were usually small but over time, they represented a considerable extra cost burden. In addition to the direct costs of corruption, this form of criminality caused significant indirect expenses to the companies in the form of delays, obstacles to competition and business activity, various extra processing and court costs, and the general insecurity of the business environment.

The greatest problems with border corruption were experienced by transport companies. During the time period covered by this study, money was collected from transport companies mainly at the border in the form of so-called excess weight fees (imposed when the total weight of the lorry was alleged to exceed a certain weight) and inside Russia in the form of various summary fines for alleged traffic violations, collected during traffic control checks. The reason why the Finnish respondents considered these fees and payments as corruption was the fact that these often had an unclear legal basis, and the companies always had to pay in cash without getting any receipt or other documentation.

The relative extra expenditure burden caused by this money collection was approximately the same for all companies. In a business sector that is subjected to stiff competition, this burden was considerable. Apart from the road transport companies, only the timber import companies had been directly subjected to border corruption.

A general policy that was followed (when it was possible) by Finnish companies in order to avoid corruption was to transfer those activities to Russian middlemen or local staff where the potential for corruption was known to exist. The companies did not want to know how these middlemen or staff members dealt with these issues. The only thing that mattered was that any problems were taken care of.

The businesspersons saw the long-term consequences of corruption in Russia as being mainly harmful for Finnish business. Corruption caused extra costs, and impaired in general terms the working environment of both foreign and local businesses. It was also considered to be harmful for the economic development of Russia, for the economic growth, and for the development of a stable state with the rule of law. Corruption was also considered to be an obstacle for Finnish companies trying to increase their trade with Russia, or to invest in the country, and also in general terms harm the business relationships between the two countries. On the other hand, some respondents argued that corruption in Russia has also created economic advantages for Finland which were, nonetheless, assessed to be of a short-term character. For instance transit

transport through Finland was thought to be significantly influenced by this circumstance.

Overall, the problems that Finnish businesses had experienced with regard to border corruption were primarily problems of road transport. The main reason seen for the problems was that the Russian operational authorities on the border and in the border region (customs, RTI³, GAI⁴, licensing authorities) either did not act at all, or acted only selectively, in accordance with international and bilateral agreements ratified by Russia, and in this way impeded the business of Finnish companies. This was usually thought to be based simply on the fact that the representatives of the authorities in strategic positions wanted to collect money illegally from the companies; sometimes, however, latent protectionist motives were also thought to be behind this behaviour. The Russian central government was assessed to be so weak that it was unable to make the border authorities comply with regulations and stop the illegal collection of money.

Russia

Of the Russian business respondents, only one said that he had been faced with corruption in Finland. Most thought that corruption in Finland is very rare, and if there is some, it is only insignificant. When asked about concrete experiences of corruption, they either said they did not know of any such cases, or they did not answer the questions at all. In this regard, the replies of the businesspersons did not illuminate the situation at all. This is likely again to be in part a consequence of the survey mode (anonymous e-mail questionnaire) applied in this case. Written replies may be seen as potentially dangerous, but also simply as requiring too much work. A face-to-face interview would most likely have provided a richer overview of the situation.

5.3 Summary and discussion

The civil servants interviewed in Finland had much less first-hand experience with corruption on the Finnish-Russian border than did the Finnish businesspersons. The business respondents were also more open than were the civil servants in speaking about corruption-related cases and problems. Many representatives of the authorities were very careful about what they said, they stated that they did not know about any cases, or they did not want to talk about them. Some of the examples that they mentioned were said to be based on hearsay only. Some civil servants also commented that Finland may not be quite as corruption-free as is commonly believed.

³ Traffic Inspection Authority

⁴ Traffic militia

None of the Finnish respondents had any experience with corruption involving the Finnish authorities. All of the corruption cases described were connected with activities of the Russian authorities or were directed at the authorities by private citizens. Some of the Finnish civil servants interviewed had first-hand knowledge of (unsuccessful) bribery attempts. These were usually bribes offered by private citizens. Usually those offering bribes had been Russians. Several civil servants said that on the basis of their observations, corruption on the Russian border and in Russia in general had decreased over recent years.

In contrast, the Finnish businesspersons had a lot to tell about the volume and variety of corruption experiences on the Russian border and its environment. The problems ranged from active grand corruption, export-related corruption, bribery of representatives of the authorities in administrative matters or in order to receive forged documents, to corruption where the businesspersons were passive victims, and to systematic money collection under a large number of pretexts. These comprised consultation services, asking for fees on the basis of ambiguous legislation, charging for documents, arbitrary charges in ad hoc situations, border crossing fees, fees collected at traffic control checks, excess weight fees, service charges, and random ad hoc money collection. Mostly, the examples concerned small-scale corruption where the businessman had the role of a victim who must consent to various extra illegal or pseudo-legal fees for permits, licences and services, or illegal or pseudo-legal fines for misdemeanours.

The greatest problems with border corruption were experienced by the Finnish road transport companies. Their main reason was thought to be that the Russian operational authorities on the border and in the border region (customs, RTI, GAI, licence authorities) were abusing their strategic positions to collect money illegally from the companies.

The Russian civil servants said they had no first-hand knowledge about corruption at the border. Most of them also stated that they had never noticed indications of corruption in their working environment. They did however point out that most corruption cases arise at points in transit at customs control, and suggested that corruption occurred in Russia with the customs authorities, in the migration service, in travel companies, at the Finnish consulate, and among Finnish entrepreneurs. Despite their lack of first-hand knowledge, they shared the opinion that in Russia, corruption is a mass phenomenon.

The replies of the Russian businesspersons did not illuminate the situation at all. They either said they did not know of any concrete corruption cases, or they did not answer these questions at all. In this respect, the applied survey mode (e-mail questionnaire) proved to be a failure.

6 COOPERATION BETWEEN THE FINNISH AND RUSSIAN AUTHORITIES

6.1 Observations of representatives of the authorities concerning cooperation between the Finnish and Russian authorities

Finland

In general, the Finnish civil servants were of the opinion that Finnish-Russian cooperation among the authorities is smooth and has improved over the years. According to them, this cooperation was not affected by corruption in any way. Some respondents said, however, that the cooperation may at times be bureaucratic and rigid. Trust was said to be the most important as a basis for good cooperation, and it was observed that years of joint efforts are required to build mutual trust. Some respondents mentioned that there are situations where one has to be careful about what one says and what information it is safe to give to the authorities of the other country. Some civil servants also commented that the exchange of information is not balanced, since one party prefers to receive rather than provide information, and it is also not possible to openly share all available information.

Some respondents were of the opinion that there is a need for more cooperation. One civil servant saw the reason for the lack of cooperation to be a reserved attitude on the part of the Russian authorities. One particular problem was felt to be that Russia had in recent times refused several requests for executive assistance made by the Finnish police, where Russian authorities were asked to interrogate crime suspects in Russia. Consequently, the volume of requests for executive assistance had decreased in both directions. (The majority of such requests come from Russia to Finland, not vice versa, and thus the respondents considered that the policy change on the part of the Russian authority was quite self-sabotaging.)

Russia

When discussing possible corruption in the cooperation between the authorities (in Finland and in Russia), the Russian civil servants pointed out that they lacked experience with such cooperation, since such functions are the responsibility of the heads of the institutions. Three respondents out of four thought that corruption can influence the cooperation between the authorities because it gives rise to distrust.

6.2 Observations of business respondents on the activities of the authorities

Finland

The Finnish business respondents were not asked about their assessment of the impact of corruption on cooperation between the Finnish and Russian authorities. Instead, they were asked how the Finnish authorities had been able to promote and defend the interests of Finnish businesses in issues concerning border corruption. They were also asked about how the information exchange with the Finnish authorities was working with regard to corruption and other problems at the border.

The representatives of transport companies were the largest sub-group among the business representatives interviewed. For this sector, also a representative of a group that promotes the interests of transport companies was interviewed. If we compare the comments made by the transport company representatives with those of the representative of their interest group, one conclusion was that information about different problems at the border was being transmitted openly and fully from the businesses to the interest group. The interest group had close contacts with the key bureaucrats. Therefore, there is no reason to suspect that the information about the problems experienced by transport companies on the border is not reaching the state authorities responsible for dealing with problems at the border. This indicates that the central authorities are well informed about the current forms of border corruption as well as of the problems that these are causing to the transport business. How this information has been utilised in the work of the authorities and in the cooperation between Finnish and Russian authorities is another question.

None of the respondents said that the Finnish authorities had had much success in this matter. However, most were of the opinion that the failure was not the fault of the Finnish authorities. The possibilities that the Finnish authorities had to influence social phenomena in Russia were seen to be practically non-existent. In individual specific problem situations, help was usually received when this was needed. In the assessments of the businesspersons, the Finnish embassy, the consulates and the other Finnish authorities operating in Russia were considered to be competent and well-informed.

The representatives of transport companies were particularly critical regarding what the Finnish authorities were doing about corruption on the border. Russian lorry companies were seen to enjoy general advantages in the traffic to Russia as a result of the corruption. Finnish lorry companies were subjected to systematic money collection on the border and on the Russian side of the border. In addition, the close connections that the Russian lorry companies were seen to have with customs and tax crimes committed on the border gave them a significant advantage in the competition for freight contracts, in

particular contracts concerning transit transport but also transport from Finland to Russia.

The Finnish authorities, both at customs and the Ministry of Transport and Communications, were said to be fully familiar with the situation. This was also confirmed in the data of the current study. Since the situation and the problems were well known, most of the transport entrepreneurs were of the opinion that the Finnish authorities had left the Finnish transport companies at the mercy of the Russians. Since the existing bilateral agreements provide instruments that could be used to influence the situation, some of the businesspersons thought that the Finnish authorities preferred their own comfort to the interests of the businesses, in that they did not want to jeopardise their friendly relationships with their Russian colleagues.

The transport entrepreneurs, on the other hand, had observed that the possibilities that the Finnish authorities had of controlling the traffic from Russia to Finland, and in this way of creating some balance in the situation, had collapsed over the last years as a result of reasons over which the authorities had no control: the decision taken in Finland on the political level to reduce police and customs staff.

The staff reduction policy has achieved small budgetary savings, at a time when thousands of jobs in the transport sector and related business have been lost and the government has lost significant tax revenue. A further factor – in addition to a lack of effort and lack of staff – influencing the inability of the Finnish authorities to control the situation was said to be the insufficient language skills of the Finnish authorities.

The Finnish transport entrepreneurs maintained that a good way to improve the situation would be the application of the principle of reciprocity: the conditions for Russian businesses in Finland should be made equivalent to the conditions for Finnish businesses in Russia. In this respect, the Finnish transport entrepreneurs felt that they had been let down by their own government and its representatives.

A constructive proposal was that the central authorities, interest groups and transport companies should create a cooperation group that would make a list of the problems in border dealings. They could then identify the three most important problems, and draft an action plan on how to eliminate or at least minimise these. It was observed matter-of-factly that, in the Russian negotiation culture, timidity or tact were not successful approaches. The notion that the negotiations could be conducted in accordance with the Finnish rules of the game was thought to be unrealistic.

The most central current problem related to border corruption was said, in the majority of the interviews, to be the discrepancy between the Russian and EU regulations concerning axle weights. This facilitates extensive illegal money collection at the border stations. The harmonisation of the Russian regulations with those in the EU was seen to be the most acute issue that Finnish authorities were expected to focus on in the Finnish-Russian cooperation

between the authorities and in the relevant bodies of the European Union. Further high-priority issues that several respondents took up were the need to reduce the number of different authorities operating on the Russian border, the importance of introducing electronic customs documents, and the need to improve the control of Russian transport of dangerous substances.

Russia

The Russian business respondents did not have anything to say about cooperation between the authorities. This is in part because they could not be interviewed in person, as this was a clear example of a topic where a lengthy face-to-face personal discussion could be able to come up with substantially more information than a written questionnaire. A serious problem in this respect was that the list of research questions designed in Finland was explicitly developed to serve as a skeleton guideline for qualitative interviews, not as a standardised questionnaire. Since the list was nevertheless used as a written questionnaire, it was particularly unsuccessful in this case where a confidential talk with enough time could have unveiled more interesting experiences and ideas than what is the case now.

6.3 Summary and discussion

In the interviews with the civil servants, cooperation between Finnish and Russian authorities was not dealt with in much detail. Mostly, the cooperation was seen to work well, and it was not seen to be influenced by corruption. Some of the Finnish civil servants, however, were not willing to discuss the Finnish-Russian cooperation between the authorities in detail. This could mean that no problems in the cooperation were perceived. But even if problems were felt to exist, the Finnish civil servants did not want to discuss them. This may be because it was important for continued cooperation that the other party is not put in a negative light. Nonetheless, some of the Finnish civil servants did experience the cooperation as working well and as having positive results.

The Finnish businesspersons were much more critical with regard to the Finnish-Russian cross-border cooperation between the authorities. The criticism was directed in particular at the weak ability of the Finnish authorities to defend and promote the interests of the Finnish businesses. According to the business respondents, the Finnish authorities had not been able to have an impact on the corruption problem in their cooperation with their Russian colleagues. It was understood that the possibilities of the Finnish authorities to influence matters were very limited. Part of the business respondents were nevertheless of the opinion that the Finnish authorities have not made full use of the instruments that were in fact at their disposal if they wished to guarantee that the Finnish and Russian companies were dealt with on an equal basis. The reason for this failure was thought to be that the Finnish authorities do not want

to jeopardise their good relationships with their Russian colleagues. However, a simple lack of effort was also seen to be an additional reason: it is easier to close one's eyes to unpleasant facts than to try to do something about them. Criticism was also expressed with respect to situations where the authorities were obviously thoroughly familiar with the problems, but kept silent about them, and even tried to silence the businesspersons who went public about the problems. The openly critical attitude of the business respondents towards the cooperation between the authorities was undoubtedly influenced by the circumstance that they were assessing the cooperation from the position of an outsider and they had no reason to be polite or muted in their criticism. Furthermore, the failures of the cooperation had a direct and concrete impact on the entrepreneurs: they paid the bill for the failures.

The Russian civil servants and businesspersons did not have much to say about these issues. This may to a great extent reflect the different data collection mode used with most of them: a written questionnaire without detailed questions is not a good instrument for such a purpose. The Russian civil servants, however, also pointed out that they lacked experience with such cooperation because such functions are the responsibility of the heads of the institutions. This conveys a clear message: if the Russian representation in the cooperation between the authorities is very much centralised and is on a high level, the remaining staff are effectively not participating, and such cooperation cannot be very pervasive.

An important comment by the majority of the Russian civil servants who were interviewed was that corruption may influence the cooperation between the authorities because it gives rise to distrust. The message is clear: trust is at the core of open and functional cooperation, and corruption and suspicions of corruption destroy this trust.

7 REGULATION AND PREVENTION OF CORRUPTION

7.1 Corruption-related training, guidelines, and familiarity with legislation

Finland

When asked directly, many of the civil servants admitted that they were not familiar with the details of the Finnish law on corruption, and even less with international anti-corruption agreements. Some respondents maintained that the moral-ethical aspect is more important than familiarity with the law, in other words the civil servant instinctively knows when his or her behaviour violates the limit of what is acceptable.

The respondents underscored that the civil servant himself or herself must know and recognise the difference between right and wrong. The civil servants thought it was almost self-evident that every civil servant is familiar with his or her rights and duties, and knows how to act accordingly.

Both the Finnish civil servants and the businesspersons were nevertheless able to define corruption rather accurately and comprehensively, and much in accordance with the Finnish Criminal Code.

A large proportion of the interviewed civil servants had not received any specific training concerning corruption or corruption prevention. They were, however, familiar with the general rule that if some indication of corruption is observed, the matter must be reported immediately.

Many civil servants had no clear recollection as to what instructions had been issued about corruption issues. Some respondents were of the opinion that some sort of training or guidelines would be necessary. On the other hand, some other respondents did not think that they needed special training. All of this would support the idea that there is a clear need for some kind of basic training regarding corruption issues.

The business respondents who were interviewed represented many kinds of different businesses, from large and medium-sized to small companies, and companies listed on the stock exchange as well as family enterprises. In most large companies, some kind of general training on corruption issues was given. In small enterprises, training and guidance on these issues had often been less systematic, or provided on an ad hoc basis. In any case, all of the respondents were familiar with the central stipulations of Finnish law. That is to say, the problem was not that the persons involved did not know what corruption is and what it is not.

Sometimes problems arose as a result of situations where it was necessary to follow different rules in Russia than those in Finland. This gives rise to a new recommendation: persons involved in Finnish-Russian trade should be given more systematic guidance concerning these issues. If such guidance is to be provided, it is important to consider what such guidance or training should comprise in order for it to be valuable for entrepreneurs. Especially for small and medium-sized enterprises about to initiate business in Russia, such training would certainly be helpful if it is practical and realistic. In point of fact, private consultancies have already been selling such services since the 1990s.

Russia

On the basis of what they said in the interviews, the Russian civil servants are fully familiar with the legislation concerning corruption. Russia has a national anti-corruption plan and several other laws concerning the topic. The civil servants knew nothing about the corresponding legislation in Finland except that some civil servants thought that in Finland civil servants who commit a corruption violation are permanently disbarred from office.

The Russian civil servants explained that for corruption to be prevented, civil servants must be familiar with the normative framework and all of the results of the efforts by the authorities regarding corruption. In addition to this, citizens must have free access to public institutions and information about the work of each institution.

7.2 The prevention of corruption

Finland

Some of the Finnish respondents said that corruption or the prevention of corruption had not played a direct role in their work, and they had not paid any particular attention to the matter. Therefore, it was also difficult to define one's own role in corruption prevention. On a general level, the civil servants made suggestions regarding various ways to prevent corruption both in Finland and in Russia. A central point regarding corruption prevention in Russia was that the salaries ought to be better in order for civil servants to secure a reasonable standard of living. If this were the case, there would be less need for corruption. According to some authors, corruption may be regarded as a part of subsistence and social security in Russia.

The Finnish respondents also said that awareness concerning corruption should be increased, and popular attitudes should be changed. In this, information has a key position. The objective should be that extra fees are not paid and that people should not think that it is possible to get privileges by paying. According to the Finnish civil servants, the rules should be clear and the punishments should be sufficiently hard. In the prevention of corruption, it is important that the authorities can be trusted and that corruption is handled in an uncompromising manner. It is important that risks are dealt with at an early stage and that when recruiting people, their reliability should have a high priority.

Suggestions for specific improvements were also made. One was the introduction of electronic customs declarations that would make double invoicing very difficult. Furthermore, making the border crossing procedure less complicated was mentioned as an anti-corruption measure. A particular problem on the Russian side of the border was observed to be that there was much bureaucracy and a large number of control points, all of which can lead to opportunities for corruption.

According to some Finnish civil servants, corruption and the abuse of privileges have already been made quite difficult in Finland by the introduction of control systems, transparency, audits and controls.

The civil servants interviewed in Finland said that one item of central importance is that the administrative system is good and reliable, and that all decisions by the authorities must be open for scrutiny both before and after the fact. It was also underscored that confidence in the authorities is a key factor in the prevention of corruption.

The civil servants often noted that parallel to surveillance and controls, the moral dimension also has a central role. Thus, it is not sufficient that the systems are good, if the people (the civil servants in particular) do not share ethically sustainable rules and they do not implement these in practice. They were of the opinion that it is central for corruption prevention that the structures of society and of the working environments facilitate and motivate individual persons to behave in a way where there is no need for corruption, and that people consider corrupt behaviour to be morally suspect. A popular notion in Finland is that one factor that prevents corruption is public shaming. If somebody is found out to have given or taken bribes, the social disapproval is considerable, and it is likely that the person in question is remembered in the future only for his or her crime rather than for his or her positive achievements. In a small country where corruption cases are rare, the incidents that are uncovered receive considerable attention in the media.

The civil servants interviewed in Finland had observed shortcomings in the prevention of corruption in both Finnish and Russian society. In Finland, there is for example a need to speak more openly about the financing of election campaigns and the many kinds of existing "old boy networks". It was also pointed out that corruption prevention should be more visible, and a credible monitoring mechanism for the UN Convention against Corruption should be created. Russia was criticised for a situation where the prevention of corruption is given much high-level attention, but on the level of concrete action, nothing of great significance takes place. In the worst case, an increase in anti-corruption resources creates more corruption, since the number of civil

servants and bureaucracy grow. Moreover, the system that allows double invoicing was heavily criticised in particular because the problem has been well known for years.

In the interviews with the Finnish business respondents, the prevention of corruption was not discussed as an independent topic. As a rule, the businesspersons thought that the Finnish authorities had no realistic possibilities of influencing Russian corruption. They also thought that the Russian central government had only very limited power to eliminate corruption.

Many respondents, in particular those who had been living in Russia for longer periods of time, thought that corruption is a permanent and unavoidable evil. Rather than being concerned about corruption prevention, they were interested in how to control corruption and to avoid the problems it caused. In this, the actions of the Finnish authorities also play a role. It was seen to be highly important that existing international and bilateral agreements are followed and that full use is made of all existing international and cross-national mechanisms in order to guarantee that the operating environment of Finnish businesses in Russia becomes the same as that provided to Russian businesses in Finland and in the European Union.

Russia

The Russian civil servants made many comments on how to prevent corruption. They said that one measure would be to make people aware of the relevant legal provisions, and to introduce strict law enforcement. They also recommended that the security services should be used to prevent violations. Regarding existing prevention mechanisms, they mentioned the commission for the observance of requirements concerning service behaviour and for settling conflicts of interest, and the commission on corruption prevention.

An important feature in the prevention of corruption was said to be increased openness. Citizens should have free access to state institutions, and they should have free access to information regarding the work of the institutions. The majority of the Russian civil servant respondents were reluctant to name any anti-corruption measures in use at the general level and in particular in their own organisation.

When asked how the civil servants saw their own role in anti-corruption work, one-half did not see that they had a role in this because if you accuse someone of corruption, you must have evidence that can be tested by law enforcement and that must hold in court. Other civil servants said that their role in corruption prevention is that they fulfil their obligations honestly and conscientiously, that they do not accept the use of information for personal gain, and that they will not take a bribe and will not allow others to take bribes.

When asked whether they had been given instructions concerning what is acceptable and what is not, four civil servants out of five said that this was the case. The remaining (one out of five) respondents said that they had never heard about such instructions. Half of them had been given instructions about the obligations related to their position, and had been presented documents on the prevention of corruption. Most of the civil servants were of the opinion that there should be special training on corruption and its prevention.

The Russian civil servants said that they are not aware of any specific examples of anti-corruption measures between the Finnish and Russian authorities, and thought that the heads of the organisations were not interested in learning about anti-corruption work in Finland. This opinion was also related to the fact that federal agencies in the regions of Russia are not allowed to make independent decisions about cooperation, and the federal central authority does not always allow them to cooperate across the border.

Regarding general methods to prevent corruption, the civil servants made a wide range of suggestions. They recommended that new staff members should be selected carefully and that the staff should be given serious anti-corruption training. Next, they recommended that there should be an increased openness in the activity of state authorities, and that there should be an open commitment to official obligations. Furthermore, they thought that complicated issues should be subjected to collective decision-making. Also, the public should have open access to state institutions.

Next, the civil servants thought that a state council should be created, the task of which would be to control the work of the state administration. They also recommended that open telephone and internet connections to institution heads be created in order to facilitate communication with them. They further said that there is a need for clear regulations and clear standards for fulfilling responsibilities.

The Russian civil servants also thought it would be helpful if all corruption cases were made public, and that those guilty should be removed from office and disbarred from official positions forever. They also maintained that improved salaries would have a positive influence on the situation.

The Russian business respondents did not appear to know much about existing anti-corruption legislation. Only a few of them said that they thought corruption is a criminal violation. The others thought that relevant legislation is only now being drafted. All of the businesspersons thought there was such legislation in Finland. None of them said that Russian companies or their own company were applying any anti-corruption measures in Finland. Regarding their own role in anti-corruption efforts, the majority said that they could inform the law enforcement authorities about corruption-related evidence.

None of the surveyed Russian businesspersons had received any corruption prevention training. All of them thought that such training would be useful. Half of them even said that such training would be essential. Not one had been given instructions on corruption issues by their employer, and they did not

know of any anti-corruption guidelines being applied in their company. A representative of a large company said that there had been anti-corruption cooperation with the state authorities, while all other respondents answered negatively to this question. None of them had any cooperation with Finnish authorities.

In discussing recommendations regarding measures for the prevention of corruption in Russia, the Russian business respondents generally thought that it is important to introduce stricter legislation in this area. One respondent also maintained that enforcement of existing legislation should be increased. Three businesspersons said that the positive experience of Finland could be used to learn about the matter.

7.3 Summary and discussion

In the Finnish data, the prevention of corruption was discussed mainly in the interviews with the civil servants. They underscored the importance of moral and ethical issues in anti-corruption work. A good knowledge of corruption-related laws was not seen to be as important as the ability to distinguish between right and wrong. This was considered to be an essential characteristic required of the authorities.

Several of the Finnish civil servant respondents admitted that they did not know the law in detail, and knew even less about international conventions on corruption. When discussing definitions, however, they were very close to that of Finnish criminal law.

The Russian civil servants said that they are fully familiar with the Russian legislation concerning corruption. Russia has a national anti-corruption plan and several other laws concerning the topic. They explained that in order for corruption to be prevented, civil servants must be familiar with the normative framework and all the efforts of the authorities regarding corruption.

The majority of the civil servants in both countries said that they had not received any kind of corruption-related training at their workplace. The respondents did nevertheless make suggestions as to how corruption could be prevented. The Finnish civil servants mentioned improved salaries (this in particular with regard to Russia), electronic customs declarations, and transparency and openness for all decision-making. The respondents maintained, however, that it is not enough that the systems are in order, if the authorities do not comply with agreed-upon ethical rules. Morality was seen as a central factor here. Corruption-related training was also seen to be useful.

Overall, the Finnish civil servants commented on these issues on a rather general level. This is understandable because they said that they had no direct experiences with corruption in their own working environment or in Finland, and it was also no acute problem in relationships with their Russian colleagues.

Consequently, they had no direct reason to think about specific corruption prevention methods.

Furthermore, the Russian civil servants made a number of suggestions as to how corruption could be reduced in Russia. Their ideas on prevention measures were not very different from those of their Finnish colleagues, but these were more specific. They were of the opinion that corruption-related training would be useful. Furthermore, they suggested that new staff should be selected carefully, the work of the authorities should be made more transparent, and that the access of the public to the authorities should be improved. They also recommended that complex decisions be subjected to collective decision-making. In their view, the normative basis of their work should be clarified, penalties for corrupt behaviour should be stiff, and persons found guilty of corruption should be disbarred from public service. Finally, they proposed that a special state council should be established with the task of controlling the work of the public administration.

The Russian business respondents also felt that training in the prevention of corruption would be important for them, and suggested that Russia could learn from the Finnish experience in anti-corruption work. They were also in favour of stiffer punishment and more effective law enforcement.

8 CORRUPTION ON THE FINNISH-RUSSIAN BORDER

This study is about corruption on the Finnish-Russian border. Because the interviewed Finnish representatives of the authorities and businesses said that they had met with corruption only on the Russian side of the border, the topic of the Finnish part of the study became in practice "Russian corruption on and near the border" as seen from the perspective of Finnish civil servants and businesspersons. The Russian part of the study corroborates this result of the Finnish study as well: the Russian civil servants and businesspersons who were surveyed had experiences with corruption issues that practically related to the Russian side of the border only.

The research method chosen provides insight into perceptions of and experiences with corruption, not binding evidence. Furthermore, **the data suffers from some bias** because of many refusals to participate in the survey in both countries.

The Finnish respondents, both the civil servants and the businesspersons, understand corruption in a rather similar fashion. This is due to two factors that are common to both groups: a general awareness of the stipulations of the Finnish Criminal Code, and personal first-hand experiences. According to the study, knowledge concerning the provisions of criminal law on corruption offences was good in both groups. Corruption was understood to comprise a broad variety of phenomena in the grey economy, as well as abuse and malpractice by the authorities. The examples described in the interviews mostly corresponded to both Finnish and Russian criminal law definitions of corruption. No indications were found of the possibility that Finnish authorities and businesspersons have essentially conflicting views about what constitutes corruption. However, a need for learning more about corruption and its regulation, including anti-corruption training, was expressed by both civil servants and businesspersons.

Of the Russian respondents, most civil servants defined corruption in strictly legal terms, of which they were well aware. The businesspersons were more pragmatic about the issue, describing circumstances where corruption could occur. All of them connected corruption with the bribing of civil servants or with civil servants blackmailing businesspersons in situations where they had the power to assist in solving their problems. In particular the businesspersons felt they would benefit from corruption-related training.

The attitude of Finnish civil servants and businesspersons towards corruption in Russia is characterised by two different aspects. First, corruption as such was not accepted. However, when operating in Russia in practice, it was often felt to be a necessary evil to yield to unlawful requests made by the Russian authorities. To engage actively in corrupting civil servants in Russia was said to be unacceptable and almost nobody described any examples of this (which

in itself indicates that this was perceived as being unacceptable). However, some business respondents surprisingly revealed that they had been guilty even of such behaviour.

Half of the interviewed Russian civil servants said that corruption cannot be justified under any circumstances. Others, however, were more flexible about this. They made reference to distress situations where they thought it is understandable if a civil servant asks for bribes. Almost without exception, the business respondents did not accept corruption. The Russian responses seemed, however, to have in mind large-scale corruption rather than the everyday small-scale collection of money. The latter was said to be created by government itself, since the existing normative regulations were conflicting and ambiguous, thus creating the opportunity for arbitrary decision-making. Many of the Finnish respondents shared this opinion as well.

A common dilemma that Finnish civil servants and businesspersons have to cope with in Russia and in cooperation with Russians is the **difference** between the two countries in the sociocultural norms related to corruption. The Finnish respondents thought that the limits of corruption in Finland are different from those existing in Russia. In cross-border activity, it was felt to be hard to decide whether one should behave according to Finnish or to Russian sociocultural norms. In specific situations, the Finnish businesspersons seemed mostly to follow the Russian sociocultural norms. Real problems arise if in such a case one also violates Russian law. Apart from the potential punishment, this also opens the possibility for extortion. The businesspersons quite commonly thought that if compliance with corruptive requests was a necessity in order to be able to operate at all, then this was a necessary evil and acceptable as such. The representatives of the authorities had a less flexible and morally uncompromising attitude.

It is interesting to notice that the Russian responses indicated also that the attitudes of Russian civil servants and businesspersons towards corruption, the "rules of the game", may be more ambiguous than many Finnish businesspersons and civil servants tend to think in terms of everyday money collection vs. "serious" corruption.

First-hand experience with corruption on the Finnish-Russian border was much more commonly reported among the Finnish businesspersons than among the Finnish civil servants. The business representatives were also more open about describing events and problems related to corruption. The corruption examples that were described were all about the behaviour of the Russian authorities or concerned situations where individual citizens had attempted to bribe a Russian or Finnish civil servant. Usually the examples concerned small-scale corruption in Russia, where the person in question agreed to pay various extra unlawful or pseudo-legal licence or service fees or unlawful or pseudo-legal fines to a representative of the Russian authorities. Even if the amounts of money involved in each case were rather small, they accumulated to form a considerable cost burden over time since the phenomenon was systematic. In addition to the direct costs of corruption, this criminality also caused

significant indirect costs to the businesses in the form of delays, obstacles to competition and business, extra processing and court trial costs, and general insecurity.

The Russian respondents said that they had no first-hand experience with corruption. On a general level, however, they did consider that Russia is a thoroughly corrupted society. Interestingly enough, as they claimed not to have direct knowledge of corruption cases, they were still able to pinpoint clearly where the vulnerabilities to corruption existed in the border environment: customs, migration authorities, travel companies, the Finnish consulate, and Finnish entrepreneurs. Here perceptions differ, as the Finnish respondents identified only one of these in an explicit sense, in that they complained about the behaviour of the Russian customs authorities. On the other hand, they identified other authorities that had caused problems for them and that were not mentioned by the Russians: the road traffic control authorities.

The most marked corruption problems on the border were experienced by Finnish road transport companies. At the time of the fieldwork of this study, money was said to be collected from them on the border in the form of so-called excess weight fees,⁵ and once in Russia, as various penalty fines⁶ collected at traffic inspection stations. The fees were collected in cash, and no receipts were given. In a strongly competitive business sector, the relative financial burden caused by this to the companies was significant. As to other business sectors, border corruption had been experienced mainly by companies importing timber to Finland.

In business sectors where it was not absolutely necessary to deal with the border authorities, the Finnish businesspersons said that they were systematically avoiding and minimising their contacts with corruption. A common method of dealing with corruption issues was to delegate the problem to Russian middlemen or local staff, when this was possible. Nobody wanted to know how these middlemen or staff members solved the problems; what was most important was that they were solved.

The Finnish businesspersons judged that the long-term consequences of Russian corruption were negative for Finnish business in Russia. Corruption caused extra costs, and it was seen to weaken the operating environment for both local and foreign businesses. This was also seen to make it difficult for Finnish businesses to increase their trade with Russia and to make investments in the country, and as being generally harmful to the economic relationship between the two countries. Some respondents pointed out that the corruption in Russia had also created economic advantages for Finland but these were seen to be of a short-term nature only. It was estimated that the current volume of

⁶ Penalty fees were said to be collected on an arbitrary basis from foreign lorries. Something can always be wrong with a lorry, and if you pay, the problem is solved.

41

⁵ "Excess weight fees" are said to be charged by road transport control authorities, indicating that these are using rigged scales and/or weighing standards that are not in line with European Union standards.

transit traffic to Russia through Finland was significantly facilitated by corruption.

Overall, the problems of Finnish companies that were related to border corruption had **mostly to do with road transport**. The basis for the unlawful collection of money was created by the fact that the Russian operational authorities on the border (customs, RTI, GAI, licence authorities) did not comply at all or complied only selectively with obligations based on international and bilateral agreements and conventions that Russia had ratified, thereby damaging the working environment of Finnish businesses. The business respondents thought that the behaviour deviating from agreed rules was mostly motivated simply by the greed of the Russian civil servants, but sometimes also by protectionist objectives. The Russian central government was assessed to be so weak that even if it wanted to, it was unable to make the border authorities behave in accordance with existing regulations and to stop the illegal collection of money.

The most central current problem related to border corruption was said, in the majority of the interviews, to be the discrepancy between the Russian and EU regulations concerning axle weights. This facilitates extensive illegal money collection at the border stations. The harmonisation of the Russian regulations with those in the EU was seen to be the most acute issue that Finnish authorities were expected to focus on in the Finnish-Russian cooperation between the authorities and in the relevant bodies of the European Union.

According to the Finnish civil servants operating at the border, the *cooperation* between the Finnish and Russian authorities functions mostly well, and corruption was not seen to have any significant impact on this. However, not all civil servants wanted to discuss problems in cooperation between the authorities. This was often likely to mean that no particular problems existed. Another reason is likely to be that it is important for continued cooperation that negative comments about the partner are avoided. The business respondents were clearly more critical with regard to the success of the crossborder cooperation between the authorities. The criticism was particularly clear regarding the ability of the Finnish authorities to defend and promote the interests of Finnish businesses in this cooperation. According to the interviewed businesspersons, the Finnish authorities had been unable to have an impact on the problem with corruption. It was accepted by most of the businesspersons that the possibilities of Finnish authorities to have impact on the problem were very limited. Some respondents were, however, of the opinion that the Finnish authorities had not made full use of the instruments that they in fact had at their disposal if they wished to create a better balance in how Finnish and Russian businesses are treated, for example, in the freight transport sector. This was believed to be the case because the Finnish authorities did not want to jeopardise their good cooperation relationships with Russians, but also because of sheer laziness: it was easier to close one's eyes to unpleasant facts than to make efforts to solve the problems. Criticism was also directed at the situation where although the authorities were

fully familiar with the problems, they kept silent about them and, in the worst case, tried to silence the businesspersons who spoke openly about the problems. The openness and the critical attitudes of the businesspersons regarding the success (or lack thereof) of cooperation between the authorities was certainly a consequence of their outsider status, since they had no reason to hide or soften their criticism. A further reason was that the lack of success in the cooperation had a direct bearing on the businesspersons because they were paying the bill for the failures.

The study provided a unique opportunity for comparing differences between Finnish and Russian perceptions and experiences with corruption. The differences were actually not very great. The Russian and Finnish civil servants seemed to have quite similar views about what was prohibited and what was acceptable. But while all of the Finnish civil servants indicated that the abuse of one's official position for personal gain was not acceptable in any kind of circumstances, only about half of the Russian civil servants shared this view. Thus, both Finnish and Russian civil servants seemed to have a similar understanding of right and wrong; the difference had to do with the issue of whether one could abuse his or her official position or not. Many Russian civil servants were of the opinion that asking for bribes was acceptable in certain kinds of situations of personal distress. Such situations could be for example where "the civil servant's family member is seriously ill and needs an expensive operation", "one's home has burned down", or "there is not enough food for the children". Thus, in the Russian responses, corruption was rather directly linked to low salaries and weak social security. In that connection, it was not condemned but understood. The abuse of one's official position in violation of regulations was also seen to be acceptable in situations where "the civil servant or his family are threatened".

Additionally, the perceptions of Russian civil servants concerning the prevalence of corruption in Russia and in Finland were very similar to those of the Finnish civil servants and businesspersons. A considerable proportion of the Russian civil servants perceived the government structures of their country to be "thoroughly corrupted", and thought that corruption in Russia was "a mass phenomenon". In their view, there was much less corruption in Finland, and the "mentality" of Finnish civil servants deviated radically from that of their Russian colleagues. On the other hand, most of the Russian civil servants maintained that it was difficult to obtain information about corruption in Russia. The problems were kept secret, and what the civil servants knew about corruption cases was based on rumours, hearsay, secondary sources and the Internet. An interesting result was also that as many as seven Russian civil servants out of ten were of the opinion that corruption had a negative impact on cooperation between Finnish and Russian authorities because it gave rise to distrust and suspicions. The self-criticism of the civil servants on the Russian side of the border was thus much more severe than what their Finnish colleagues said about them. On the other hand, the Russian civil servants made equally positive assessments about their Finnish colleagues as those made by the Finnish civil servants with regard to the Russians.

Overall, the results of the study concerning corruption on the Finnish-Russian border and export trade from and through Finland to Russia were quite similar to expectations based on earlier research results, and reinforce the perception about its institutionalised nature. What may perhaps be surprising is that the corruption-related perceptions of the Finnish civil servants and businesspersons were so similar, and maybe even more surprising is that they were overall quite close to the perceptions of the Russian civil servants. A further surprise was perhaps that individual Russian civil servants discussed the phenomenon quite openly on a general level, but first-hand experience or incidents in their own working environment were topics that the Russian civil servants – as was the case with their Finnish colleagues – were usually not willing to discuss.

The Finnish civil servants and businesspersons seemed to share the understanding that bribes are criminalised both in Russia and in Finland. However, many of them expressed the view that they would **benefit from anti-corruption training** and some also thought it would be good to learn more about the Russian system. It was noted that persons involved in Finnish-Russian trade should be given more systematic guidance concerning Russian law and sociocultural norms. This would be valuable especially for small and medium-sized enterprises about to initiate business in Russia. In point of fact, private consultancies have already been selling such services since the 1990s.

The Russian civil servants and businesspersons expressed more uncertainty and many of them thought that anti-corruption training would be welcomed. They also felt that it would be useful to learn more about the Finnish system and corruption prevention in Finland.

The information exchange between the Finnish entrepreneurs and authorities seems to be working well. The Finnish authorities are usually well informed about problems troubling Finnish businesspersons on the Russian border. The feedback from the authorities to businesses does not, however, seem to be functioning properly. The way in which some authorities have attempted to deny or trivialise problems in public has probably been harmful rather than constructive with regard to attempts to solve the problems.

In work for solutions to border corruption problems, more **open and active cooperation and information exchange between Finnish authorities and businesses** would probably be beneficial. Such cooperation would increase the confidence of the businesses in the authorities of their own country and would diminish unnecessary suspicions. It might, however, also help the authorities by opening new perspectives on the problems and ways of solving them. How to organise the cooperation in practice is a separate issue. As such, the Finnish authorities have quite limited possibilities for influencing the corruption situation on the Russian side of the border. The criticism directed at the current policy should not, however, be shrugged off. The Finnish companies operating in Russia and having daily contact with the border have first-hand experience with the situation, and are familiar with local circumstances, problems and their solutions. This makes them a valuable resource that the authorities should use more than they are currently doing. A concrete example was given in the

majority of the interviews with Finnish businesspersons: the discrepancy between the Russian and EU regulations concerning axle weights facilitates extensive illegal money collection at the border stations, and the Finnish businesspersons expect that this issue is dealt with in Finnish-Russian authority cooperation.

Likewise, the Russian report includes recommendations on how to achieve change for better in the situation. First of all, it recommends that special training should be organised in government agencies with corruption prevention as the central topic. Furthermore, the work of the public authorities must be made more transparent and they must be made more accessible to the general public. Next, businesses need to get organised in order to decrease their dependence on arbitrary decisions by bureaucrats. The **influence** of civil servants on the allocation of funds and benefits must be decreased. Authorities must introduce a strict selection of staff, and the higher-level officials need to work with their staff constantly on these issues. Public **councils** should be established to control the activity of the administration. Institution heads must be accessible, and this can be achieved by providing open internet and telephone connections to them. The incoherent normative base that facilitates arbitrary decisions and corruption must be made less ambiguous so that it is simple and clear. Corruption cases must be made public, and the civil servants who have made themselves guilty of corruption must be removed from office and disbarred from public service. Finally, the civil servants should be paid decent salaries, and a system should be established of granting benefits for those who disclose evidence of corruption.

In summary, both the Finnish and the Russian respondents had quite similar views concerning the actual corruption situation on the border, its causes and consequences. Their recommendations as to what should be done to improve the situation were also often similar. Thus, the problem seems not to be that Finns and Russians perceive corruption in a fundamentally different way (although there seem to be differences also in this respect), or that there are no effective means to eliminate corruption from the border. The main problem would rather seem to be in the lack of political will to put these means into practice, both in Russia and in Finland.

Literature utilised in the original Finnish and Russian country reports

- Alvazzi del Frate, Anna & van Kesteren, John (2004). Criminal Victimisation in Urban Europe. Key Findings of the 2000 International Crime Victim Surveys. UNICRI.
- Aromaa, Kauko & Junninen, Mika (2000). Corruption in Road Transports. XIII Baltic Criminological Seminar, Tallinn, 24–27 June 2000. Unpublished paper.
- Aromaa, Kauko & Lehti, Martti (2001). Pietari suomalaisyritysten turvallisuusympäristönä 1994–1999. Helsinki: Oikeuspoliittisen tutkimuslaitoksen julkaisuja 176. [Roofs and the security of Finnish Companies in Russia and the Baltic Countries 1995–2000. National Research Institute of Legal Policy, Publication No. 176. Helsinki.]
- van Dijk, Jan, van Kesteren, John & Smit Paul (2007). Criminal Victimisation in International Perspective. Key Findings from the 2004–2005 ICVS and EU ICS. Tilburg University, UNICRI, UNODC.
- European Commission (2008). The attitudes of Europeans towards corruption. Special Eurobarometer 291.
- Generalov, Outi-Elina (2008). Kaksoislaskutus Venäjän kaupassa. Opinnäytetyö. Jyväskylän ammattikorkeakoulu. [Double Invoicing in Russian Trade. Thesis study. Jyväskylä University of Applied Sciences.]
- Global Corruption Barometer 2009. Transparency International.
- Harjutsalo, Linda (2008). Virkarikokset: Erityisesti lahjonta virkarikoksena. Julkaisematon pro gradu -tutkielma. Vaasan yliopiston hallintotieteiden tiedekunta. [Offences in office: bribery as an offence in office. Unpublished Master's Thesis. University of Vaasa, Faculty of Administrative Sciences.]
- Hallituksen esitys Eduskunnalle korruption vastaisen Yhdistyneiden Kansakuntien yleissopimuksen hyväksymisestä sekä laeiksi sopimuksen lainsäädännön alaan kuuluvien määräysten voimaansaattamisesta ja rikoslain 30 luvun 12 §:n muuttamisesta (HE 197/2005). [Government Bill to Parliament concerning approval of the United Nations Convention against Corruption; the enactment of regulations under the Convention legislation; and amendments to Chapter 30, section 12, of the Finnish Penal Code.]
- Henttonen, Maija (2007). Venäjän tullin toiminta hyvän hallinnon periaatteiden valossa. Julkaisematon pro gradu -tutkielma. Tampereen yliopiston johtamistieteiden laitos. [Russian customs operations in the light of the Principles of Good Governance. Unpublished Master's Thesis. University of Tampere, Department of Management Studies.]
- Heusala, Anna-Liisa, Lohiniva, Anja, Malmi, Antti (2008). Samalla puolella eri puolella rajaa. Rajaturvallisuuden edistäminen Suomen ja Venäjän viranomaisyhteistyönä. Tampere: Poliisiammattikorkeakoulun tutkimuksia 30/2008. Espoo: Raja- ja merivartiokoulun julkaisusarja 1. Tutkimuksia no. 2. [On the same side, yet opposite sides of the border. Promoting border security through Finnish–Russian authority cooperation. Tampere: Police

- College of Finland, Research Series 30/2008. Espoo: Border and Coast Guard Academy, Publication Series No.1. Research Series No. 2.]
- Isaksson, Paavo (1997). Korruptio ja julkinen valta. Tampereen yliopiston yhteiskuntatieteiden tutkimuslaitoksen julkaisuja 15/1997. [Corruption and Public Authority. Research Reports of the Research Institute of the Social Sciences, University of Tampere, Series 15/1997.]
- Jukanen, Janne (2007). Kuljetusyritysten kilpailukyvyn ongelmatekijöiden analyysi Venäjän ja Suomen välisissä maantiekuljetuksissa. TKK, Tuotantotalouden laitos. Diplomityö teollisuustaloudessa. [Analysis of key factors affecting the competitiveness of transport companies in road transport between Russia and Finland. Helsinki University of Technology, Department of Industrial Engineering and Management. Master's Thesis.]
- Korruptiotilannekuva 2008. Vantaa: Keskusrikospoliisi. [Corruption Status Report 2008. Vantaa: National Bureau of Investigation.]
- Korruptiotilannekuva 2009. Jenni Klemola. Vantaa: Keskusrikospoliisi. [Corruption Status Report 2009. Jenni Klemola. Vantaa: National Bureau of Investigation.]
- Laitinen, Ahti (2004). Suomalainen poliisi ja korruptio. Oikeus (33):1, 80–87. [Finnish Police and Corruption. Oikeus (Journal of the Finnish Society for Law and Social Sciences) (33):1, p. 80–87.]
- Larjavaara, Ilmari (1999). The Russian administration: a study of Russian authorities encountering international road freight transport. Helsinki: Finnish Institute for Russian and East European Studies.
- Larjavaara, Ilmari (2007). Funktionaalinen transitioteoria. Hallinto, oikeusjärjestys ja instituutiot Venäjällä. Jyväskylä: Kikimora. [Functional transition theory: administration, legal order and institutions in Russia. Jyväskylä: Kikimora.]
- Lehtola, Markku & Paksula, Kauko (1997). Talousrikosten tilannetorjunta. Helsinki: Oikeuspoliittisen tutkimuslaitoksen julkaisuja 142. [Situational Crime Prevention and Economic Crime. Helsinki: National Research Institute of Legal Policy, Publications, No. 142.]
- Leppälä, Jussi (2009). Tulliselvitysrikos. Lainsäädäntöehdotuksen arviointia. Tampere: Poliisiammattikorkeakoulun raportteja 79/2009. [Customs clearance offences. An assessment of proposed legislation. Tampere: Police College of Finland, Report Series No. 79/2009.]
- Lohiniva, Anja (2008). "Mistä se oikea partneri löytyy?" Selvitys suomalaisvenäläisestä viranomaisyhteistyöstä talousrikosten torjunnassa ja tutkinnassa Suomen keskusrikospoliisin näkökulma. Tampere:
 - Poliisiammattikorkeakoulun raportteja 70/2008. [Finding the Right Partner: A study of Finnish-Russian authority cooperation in combating and investigating economic crime from the National Bureau of Investigation of Finland's perspective. Tampere: Police College of Finland, Report Series No. 70/2008.]
- Marttila, Juhani (2007). Venäjä Suomen puurakennusteollisuuden mahdollisuutena. Metsä- ja puuteknologian pro gradu -työ. Joensuun yliopiston metsätieteellinen tiedekunta. [Russia as an opportunity to the

- Finnish wood construction industry. Master's Thesis. University of Joensuu, Faculty of Forest Sciences.]
- Muzalevskaja E.A. (2006). Corruption in the system of civil service in Russia: sources and tendencies 1992–2005.
- Ollus, Simon-Erik (2005). Law and Order: Some Impacts of Introducing a Law Enforcement Mechanism on Growth in Predatory Economies. Unpublished Master's Thesis. University of Helsinki, Department of Economics.
- Pihlajamäki, Antti (2006). Julkisen sektorin lahjonta Suomen rikoslaissa ja korruptionvastaiset kansainväliset sopimukset. Rikosoikeudellisia kirjoituksia. Suomalainen lakimiesyhdistys 8/2006, 405–419. [Public sector bribery in the Finnish Penal Code and international agreements against corruption. Finnish Lawyers' Society journal, 8/2006, p. 405–419.]
- Pihlajamäki, Antti (2008). Yksityisen sektorin lahjonta. [Bribery in the Public Sector.] Defensor Legis 4/2008, 516–520.
- Poliisibarometri 2007. Poliisin ylijohdon julkaisusarja 4/2008. [Police Barometer 2007. Supreme Police Command of Finland, Publication Series No. 4/2008.]
- Proshina, E.M. (2000). Public education and its role in public influence in the process of corruption prevention. Thesis and reports, 27–28 April. St. Petersburgh.
- Rasimus, Salme (2007). Korruption ja alhaisen talouskasvun välinen noidankehä. Julkaisematon pro gradu –tutkielma. Helsingin yliopiston kansantaloustieteen laitos.
 - [The vicious circle of corruption and slow economic growth. Unpublished Master's Thesis. University of Helsinki, Department of Economics.]
- Rikollisuustilanne 2007. Rikollisuus- ja seuraamusjärjestelmä tilastojen valossa. Helsinki: Oikeuspoliittinen tutkimuslaitoksen tutkimuksia 238. [2007 Annual Crime Report, Crime and Criminal Justice in Finland. Helsinki: National Research Institute of Legal Policy, Publication No. 238.]
- Salminen, Ari, Viinamäki, Olli-Pekka, Ikola-Norrbacka, Rinna (2007). The Control of Corruption in Finland. Administration and Public Management Review 9/2007. Bucharest: Academy of Economic Studies.
- Salminen, Ari & Ikola-Norrbacka, Rinna (2009). Trust and Integrity Violations in Finnish Public Administration: The Views of Citizens. Halduskultuur (10):1, 74–93.
- Sisäasiainministeriö 2008. Turvallinen elämä jokaiselle. Sisäisen turvallisuuden ohjelma. Sisäasiainministeriön julkaisuja 16/2008. [Ministry of the Interior of Finland. Safety first Internal Security Programme. Publications of the Ministry of the Interior 16/2008.]
- Sungurov A. Ju. (2000) (ed.). Public initiatives and corruption prevention. St. Petersburgh: Norma.
- Taybakov, A.A., Demin, P.I., Tumanov, R.V., Klemeshov, P.A. (2009). Corruption on Russian-Finnish border. Comparative analysis in Finland and the Republic of Karelia, Russia. The Northern (Petrozavodsk) Branch of the Russian Legal Academy of the Ministry of Justice of the Russian Federation. Unpublished manuscript.

Transparency International, www.transparency.org, downloaded 16 June 2009. Transparency International, Finland website:

http://www.transparency.fi/, downloaded 3 April 2009;

http://www.transparency.fi/Documents/tietoa_korruptiosta/, downloaded 1 July 2009.

Ulkoasiainministeriö 2002. Korruption vastaisen toiminnan käsikirja. Helsinki:UM.

[A handbook of anti-corruption techniques for use in international development cooperation. Ministry for Foreign Affairs, Helsinki, 2002.]

Viljanen, Pekka (2008). Yritysten ja virkamiesten välinen kanssakäyminen rikoslain 16 ja 40 lukujen lahjusrikossäännösten kannalta – sallitun ja kielletyn välistä rajanvetoa erityisesti oikeuskäytännön valossa [Relations between companies and authorities with respect to the bribery offence regulations prescribed in Chapters 16 and 40 of the Finnish Penal Code – the distinction between what is permitted and prohibited, particularly in the light of legal practice]. Defensor Legis 4/2008, 500–515.

Viuhko, Minna, Lehti, Martti (2009). Korruptio Suomen ja Venäjän rajalla. Suomalaisten viranomaisten edustajien ja liikemiesten kokemuksia ja näkemyksiä korruptiosta itärajalla. Oikeuspoliittisen tutkimuslaitoksen tutkimustiedonantoja 99. HEUNI Publication Series No. 60. [Corruption on the Finnish-Russian Border. Experiences and views of Finnish authority staff and businessmen of corruption on Finland's eastern border. National Research Institute of Legal Policy. Research Communication No. 99. HEUNI Publication Series No. 60.]

Weurlander, Matts (2007). Korruptionsdiskursen i landet utan korruption. Julkaisematon pro gradu-tutkielma. Helsingin yliopiston sosiologian laitos. [Corruption Discourse in the Country without Corruption Unpublished Master's Thesis. University of Helsinki, Department of Sociology.]